Author’s response to reviews

Title: Obstacles to recruitment in paediatric studies focusing on mental health in a physical health context: The experiences of clinical gatekeepers in an observational cohort study

Authors:

Maria Loades (m.e.loades@bath.ac.uk)
Lucie Smith (las90@bath.ac.uk)
Nina Higson-Sweeney (nina2.higson-sweeney@live.uwe.ac.uk)
Lucy Beasant (lucy.beasant@bristol.ac.uk)
Paul Stallard (p.stallard@bath.ac.uk)
David Kessler (david.kessler@bristol.ac.uk)
Esther Crawley (ester.crawley@bristol.ac.uk)

Version: 1 Date: 04 Feb 2019

Author’s response to reviews:

Please find out amendments detailed as a cover letter, with tracked changes in our revised manuscript.